

Format for ANSWERING REVIEWERS



June 4, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 3531-review.doc).

Title: Biomolecular basis of the role of diabetes mellitus in osteoporosis and bone fractures

Author: Bipradas Roy

Name of Journal: *World Journal of Diabetes*

ESPS Manuscript NO: 3531

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2. Revision has been made according to the suggestions of the reviewer

3. References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Diabetes*.

Sincerely yours,

A handwritten signature in black ink that reads 'Bipradas Roy'.

Bipradas Roy

B.Sc. in Biotechnology and Genetic Engineering

Khulna University

Khulna-9208

Bangladesh

Cell: +8801737260794

E-mail: bplobbge06ku@gmail.com